Author's response to reviews

Title: Medium and longterm adherence to postabortion contraception among women having experienced unsafe induced abortion in Dar es Salaam

Authors:

Vibeke Rasch (v.rasch@pubhealth.ku.dk)
Fortunata Yambesi (foetyy@yahoo.com)
Siriel Massawe (smassawe@muchs.ac.tz)

Version: 4 Date: 17 June 2008

Author's response to reviews:

- The authors' qualifications have been removed from the manuscript
- Titles have been removed from the list of authors
- The email addresses and the affiliations have been separated and each email address is preceded by the author's initials
- All unnecessary capitals have been removed from the section headings
- Keywords and Word-counts have been removed
- Author contributions has been changed to Authors' contributions